Author's response to reviews

Title: Factors associated with commencing smoking in 12-year-old students in Catalonia (Spain): a cross-sectional population-based study

Authors:

Empar Valdivieso (evaldivieso.tarte.ics@gencat.cat)
Cristina Rey (crey.tarte.ics@gencat.cat)
Marisa Barrera (mbarrera.tarte.ics@gencat.cat)
Victoria Arija (victoria.arija@urv.cat)
Josep Basora (jbasora.tarte.ics@gencat.cat)
Josep R Marsal (jmarsal.lleida.ics@gencat.cat)
Study Group TAB_ES (tabes2007@gmail.com)

Version: 4 Date: 29 October 2010

Author's response to reviews:

Title : Factors associated with commencing smoking in 12-year-old students in Catalonia (Spain): a cross-sectional population-based study
Journal: BMC Public Health
MS ID : 1400509144369000

Thanks for reviewing our manuscript. We checked the manuscript carefully and made all the changes (major and minor revision) that the BMC Editorial Production Team suggested us.

Best wishes,
Empar Valdivieso
Corresponding author
Email:evaldivieso.tarte.ics@gencat.cat